To whom it may concern,

Thank you for considering our manuscript for publication in the World Journal of Radiology. I appreciate your feedback and have incorporated the following revisions.

(1) The “Author Contributions” section is missing.
   • The Authors Contributions section is now included

(2) The authors did not provide original pictures. Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor
   • I provided a word document including the 3 tables, correctly formatted
   • I provided a powerpoint document including the 2 graphs, in SVG format

(3) The “Article Highlights” section is missing.
   • I added the Article Highlights at the end of the main text.

Please let me know if there is anything else is needed.

Best regards,
Peyton Cramer, MD